



# **Geophysical Research Letters**

## **RESEARCH LETTER**

10.1029/2018GL080959

### **Key Points:**

- Anthropogenic climate change is projected to enhance fire weather across most burnable global land surfaces during the twenty-first century
- Emergence of fire weather conditions from natural variability is modeled to occur in the first half of the twenty-first century in many regions
- Extent of fire-weather emergence is twice as large when global temperature surpasses 3 °C, compared to 2 °C, above preindustrial levels

### Supporting Information:

Supporting Information S1

### Correspondence to:

J. T. Abatzoglou, jabatzoglou@uidaho.edu

### Citation:

Abatzoglou, J. T., Williams, A. P., & Barbero, R. (2019). Global emergence of anthropogenic climate change in fire weather indices. *Geophysical Research Letters*, 46, 326–336. https://doi.org/ 10.1029/2018GL080959

Received 22 OCT 2018 Accepted 26 NOV 2018 Accepted article online 30 NOV 2018 Published online 8 JAN 2019

#### ©2018. American Geophysical Union. All Rights Reserved.

## **Global Emergence of Anthropogenic Climate Change in Fire** Weather Indices

John T. Abatzoglou<sup>1</sup>, A. Park Williams<sup>2</sup>, and Renaud Barbero<sup>3</sup>

<sup>1</sup>Department of Geography, University of Idaho, Moscow, ID, USA, <sup>2</sup>Lamont-Doherty Earth Observatory of Columbia University, New York, NY, USA, <sup>3</sup>Irstea, Mediterranean Ecosystems and Risks, Aix-en-Provence, France

**Abstract** Changes in global fire activity are influenced by a multitude of factors including land-cover change, policies, and climatic conditions. This study uses 17 climate models to evaluate when changes in fire weather, as realized through the Fire Weather Index, emerge from the expected range of internal variability due to anthropogenic climate change using the time of emergence framework. Anthropogenic increases in extreme Fire Weather Index days emerge for 22% of burnable land area globally by 2019, including much of the Mediterranean and the Amazon. By the midtwenty-first century, emergence among the different Fire Weather Index metrics occurs for 33–62% of burnable lands. Emergence of heightened fire weather becomes more widespread as a function of global temperature change. At 2 °C above preindustrial levels, the area of emergence is half that for 3 °C. These results highlight increases in fire weather conditions with human-caused climate change and incentivize local adaptation efforts to limit detrimental fire impacts.

**Plain Language Summary** Observed increases in the frequency and severity of fire weather have been observed across portions of the globe over the past half century. We used climate models to identify where and when anthropogenic climate change causes fire weather conditions to exceed that of natural variability. Modeling results show that emergence for some fire weather indices is already under way for a sizable portion of the globe, including much of southern Europe and the Amazon, and with an expansion of this area with continued warming over the twenty-first century. These findings suggest substantial increases in fire potential in regions where vegetation abundance and ignitions are not limiting, highlighting the urgency to adapt to changes in fire disturbances and hazards.

## **1. Introduction**

Fire is both a natural ecological process and tool used by humans that impacts ecosystems, water supplies, carbon budgets, and society (Bowman et al., 2017; Johnston et al., 2012; Randerson et al., 2012; Turetsky et al., 2015). Retrospective estimates of burned area gleaned through proxy records and modeling support a global decline since the industrial revolution due to societal factors such as land use conversion and fire suppression (Andela et al., 2017; Arora & Melton, 2018; Marlon et al., 2012). However, increases in broad-scale burned area are well documented in some portions of the globe over the past half century (Kasischke & Turetsky, 2006; Westerling, 2016), suggestive of the influence of changing climate conditions and increased temperatures on fire activity when other ingredients such as biomass abundance are not limiting. Climate variability imparts distinct relationships on fire activity mediated through its influence on biomass production and fuel aridity (Bradstock, 2010), with fuel aridity exhibiting strong concurrent relationships with burned area in flammability-limited regions, while strong antecedent relationships are observed between pluvial conditions one to two years prior to the fire season and burned area in fuel-limited regions (Abatzoglou et al., 2018; van der van der Werf et al., 2008).

Proxies for fuel aridity, including fire weather indices, indicate positive (drying) trends across much of the globe over the past four decades (Jolly et al., 2015). In western United States forests, approximately half of the observed increase across a suite of fuel aridity proxies from 1979 to 2015 was attributed to anthropogenic warming (Abatzoglou & Williams, 2016), and Gillett et al. (2004) noted a detectable influence of anthropogenic climate change on increased summer temperatures and burned area across Canadian forests. Detection and attribution of global fire activity to anthropogenic climate change is confounded by influences of other anthropogenic activities such as land-cover change, population, and fire suppression as well as temporally limited satellite-based fire records. Studies have examined the relative influences of these factors to observed and projected global fire activity (Knorr et al., 2016; Pechony & Shindell, 2010; Ward et al., 2018),

but have not explicitly tried to identify when and where the influences of climate change will become perceivable locally, regionally, or globally. Joint attribution efforts for natural systems are more challenging than attribution of meteorological and climate variables to anthropogenic climate change (Rosenzweig et al., 2008), let alone coupled natural-human processes such as fire. As an alternative, a limited set of attribution studies that focus on atmospheric drivers and enablers of fire have been conducted but have often been limited geographically and methodologically (e.g., Abatzoglou & Williams, 2016; Gillett et al., 2004; Kirchmeier-Young et al., 2017; Tan et al., 2018).

Attribution studies have elucidated the influence of anthropogenic climate change across a host of metrics such as temperature, precipitation, ocean temperatures, and mountain snowpack (e.g., Barnett et al., 2008; Pierce et al., 2008) as well as distinct events (e.g., Risser & Wehner, 2017; Williams et al., 2015). These studies interrogate the relative contribution of causal factors to changes in some measurable element or the magnitude of frequency of a distinct event to a given forcing versus that which arises solely due to internal climate variability (Stott et al., 2015). An alternative approach that complements traditional attribution methods is the framework of the time of emergence (ToE) of anthropogenic climate change. ToE is defined when a signal forced by anthropogenic climate change exceeds some measure of the internal climate variability that occurs in the absence of anthropogenic forcing. Studies have demonstrated the ToE for first-order climate variables like mean temperature (Diffenbaugh & Scherer, 2011; Hawkins & Sutton, 2012), precipitation and temperature extremes (King et al., 2015; Tan et al., 2018), and sea level rise (Lyu et al., 2014). The signal-to-noise framework of ToE can not only highlight potential timelines for adaptation measures but also may be particularly important for ecosystem processes including fire, given that the approach queries whether changes exceed the natural variability upon which many systems are adapted to.

This study uses the ToE framework to address when and where the signal of anthropogenic climate change emerges from internal climate variability through the lens of the Fire Weather Index (FWI). The FWI from the Canadian Forest Fire Weather Index System is considered in this study given its widespread usage globally (Di Giuseppe et al., 2016; Field et al., 2016; Groot et al., 2014); its reflection of potential fire intensity tied to fuel aridity and fire weather irrespective of land cover and biomass; its sensitivities to temperature, precipitation, humidity, and wind speed (e.g., Flannigan et al., 2016); and its established empirical relationships to burned area across broad regions of the globe (e.g., Abatzoglou et al., 2018). While previous studies have shown changes in global fire weather metrics such as the FWI under anthropogenic climate change (e.g., Flannigan et al., 2013), these have been limited to a small number of models, and have not explicitly demonstrated if or when anthropogenic-forced changes emerge from internal variability. We aim to address this knowledge gap in projected changes in fire weather using a large set of models and address differences both geographically and across a set of different measures of FWI.

## 2. Data and Methods

Simulations from 17 models participating in the Fifth Phase of the Coupled Model Intercomparison Project were used in this analysis (Table S1). Models were selected that archived daily maximum temperature (*tas-max*), accumulated precipitation (*pr*), mean wind speed (*sfcWind*), and daily minimum relative humidity (*rhsmin*) for historical (1861–2005) and future (2006–2099) experiments, with the latter using Representative Concentration Pathway 8.5. Analyses were limited to a single realization (r11p1) per model. Ten models lacked archived *rhsmin* prior to 1950 (Table S1). For these models, we extended *rhsmin* back to 1861 using *tasmax* and daily mean 2-m specific humidity and bias correction procedures (Supporting information). All model output was regridded to a common 2.5° grid from 50°S to 70°N. Land areas where more than 80% of current land cover was classified as water, snow and ice, or barren or sparsely vegetated by MODIS land cover type product (Friedl et al., 2010) were considered unburnable and masked out. Results are presented using raw output from global climate models (GCMs). Experiments using univariate and multivariate bias correction procedures showed qualitatively similar conclusions (Cannon, 2018; Li et al., 2010; Pierce et al., 2015; supporting information).

The FWI is an output of the Canadian Forest Fire Danger Rating System (Van Wagner, 1987) that is a widely used numerical indicator of potential fire intensity. The system was originally developed as a numerical proxy for fire behavior in evergreen forests of eastern Canada, but has been shown to be an effective correlate of intraseasonal to interannual fire activity globally across diverse vegetation types, most notably in

flammability-limited environments (Abatzoglou et al., 2018). The FWI integrates the effects of fuel availability through a buildup index that combines three different classes of fuel moisture with the potential rate of fire spread. The former is a climate-weather hybrid that entrain information (e.g., precipitation, temperature) on time scales ranging from the previous day to the preceding couple months, whereas the latter is predominantly influenced by daily fluctuations in weather (e.g., wind speed). Calculation of FWI requires daily 24-hr accumulated precipitation as well as temperature, humidity, and wind speed at 1200 local time. Given the barriers in obtaining observed and modeled data at 1200 local time, we use daily maximum temperature (*tas-max*), daily minimum relative humidity (*rhsmin*), daily accumulated precipitation (*pr*), and daily mean wind speed (*sfcWind*) following prior global analyses (e.g., Abatzoglou et al., 2018; Jolly et al., 2015).

We chose four annualized FWI metrics as indicators of fire potential including two frequency-based metrics that examine the number of days per year of extreme fire weather and length of the fire weather season, and two magnitude-based metrics that examine annual maxima and seasonal averages in FWI. These metrics were selected to highlight differences in ToE across different statistical properties of FWI that have established relationships to fire activity. First, we calculated for each pixel the number of days per year of high FWI, quantified as days exceeding the 95th percentile (FWI<sub>95d</sub>) because fire growth preferentially occurs during high fire danger periods (e.g., Barbero et al., 2014). Second, we calculated the fire weather season length (FWI<sub>fwsI</sub>) following Jolly et al. (2015), defined as days per year exceeding the midrange point. Third, we calculated the annual maximum FWI (FWI<sub>max</sub>) which provides a measure of extreme fire risk (Tan, Chen, & Gan, 2018). Finally, we calculated the annual peak 90-day mean FWI (FWI<sub>fs</sub>) as a measure of seasonal fuel aridity during the primary fire season—as studies have shown that a majority of annual burned area at regional scales occurs within three consecutive months and interannual burned area variability exhibits significant relationships to concurrent FWI (e.g., Abatzoglou et al., 2018). A flexible 90-day window was used to calculate FWI\_{fs} as it provided slightly more explained variance to interannual burned area records than fixed dates in prior studies (Higuera et al., 2015), and may account for changes in phenology of the fire season.

A variety of approaches have been used in ToE studies leading to challenges in cross-study comparisons. These include various baseline periods and approaches for estimating internal climate variability, statistical procedures for guantifying emergence, and uses of model ensembles (Diffenbaugh & Scherer, 2011; Hawkins & Sutton, 2012; Lehner et al., 2017). We used a 50-year baseline period from 1861 to 1910 when anthropogenic forcing was closer to a preindustrial state to gauge quasi-natural variability (e.g., King et al., 2015). This 50-year period was also used to define the local 95th percentile threshold value for calculating FWl<sub>95d</sub> and midpoint range (average of maximum and minimum value of FWI) for calculating FWI<sub>fwsI</sub>. Emergence was quantified using a signal-to-noise criterion (e.g., Hawkins & Sutton, 2012). This was done using moving 30-year windows starting in the year 1980 on the basis of the signal of change, defined as the difference of means between a given 30-year period and the baseline period, exceeding the noise of natural variability, defined herein as one standard deviation (calculated for the baseline period). We additionally calculated ToE using a right-tailed t test (p < 0.05). This alternative approach showed more widespread and earlier emergence than the approach we adopted; thus, we consider the signal-to-noise approach used here to be more conservative. To limit pseudo-emergence, ToE was indexed as the first year in a moving 30-year window (e.g., 2011 was the ToE for 30-year period from 2011 to 2040; e.g., King et al., 2015), whereby criteria are met for all subsequent moving 30-year periods. Additionally, we restrict emergence to occurring no later than 2050 to ensure persistence for at least 2 decades after initial detection, following previous methods (Hawkins et al., 2014; King et al., 2015).

Estimates of local ToE were conducted at the 2.5° grid of interpolated GCM output as well as at larger, regional scales. Previous studies have shown that ToE is later at local scales than regional scales (Maraun, 2013), as internal variability is reduced through geographic aggregation. Local emergence of FWI may be ecologically important for fire regimes and trajectories of species distributions. However, regional emergence may be informative for broader adaptation efforts as widespread fuel aridity is associated with synchronous regional fire activity in flammability-limited fire regimes (Abatzoglou & Williams, 2016). Likewise, geographic synchrony in fire weather indices has direct implications for regional fire suppression efforts and broad-scale implications for smoke-related health impacts. We aggregate data to IPCC SREX regions as used in previous efforts (King et al., 2015) for assessing regional emergence of FWI. Aggregation was performed by calculating averages of daily FWI for all pixels within the bounds of each region and summarizing the four metrics as done at the pixel scale.



**Figure 1.** Time of emergence of anthropogenic climate change for (a) the frequency of days exceeding the 95th percentile, (b) the length of the fire weather season, (c) the peak 90-day average FWI, and (d) annual maximum FWI. Mapped is the 17-model median date at which the anthropogenic signal emerges based on the change exceeding the standard deviation of the baseline period. Areas of light gray highlight where the signal does not emerge for at least eight models by 2050. Unburnable lands where >80% of the area is water, snow, and ice or barren or sparsely vegetated are shown in dark gray. The percent of burnable land with emergence by 2050 is denoted in the bottom left corner of each map.

Given the potential challenges with reconciling varied climate sensitivity of individual models and uncertainty regarding emission pathways, we consider an alternative to the ToE concept by replacing time with global mean temperature following Hawkins and Sutton (2012). We refer to this as the Global Temperature of Emergence (GToE), which uses the same framework as ToE, but may be advantageous because it is agnostic with respect to model and emission pathway and is more directly aligned with climate policy targets. The GToE differs from ToE in that GToE is defined by the 30-year average global mean air temperature difference for the period when ToE is first identified relative to baseline conditions for 1861–1910 (i.e., quasi pre-industrial conditions). Identifying GToE for FWI metrics may provide insight into risk avoidance though mitigation efforts without explicitly comparing differences in climate forcing by inferring differences in model-derived transient climate response given the varied equilibrium climate sensitivity of the models used (between 2.1 and 4.7 K; Table S1). However, this approach is conservative given the noted differences between transient and equilibrium climate responses.

Emergence was tallied separately for each model. We focus on multimodel median ToE and GToE in the results section. Emergence is thus stated to not occur where fewer than 9 of the 17 models show emergence. When a majority of the GCMs show emergence, ToE is reported as the year when the ninth model emerges from internal variability. Results of regional emergence for individual models are provided in the supporting information to highlight model uncertainty in ToE.

## 3. Results

Local emergence of the four FWI metrics by 2050 was found for between 33 and 62% of global burnable land surfaces (Figure 1). Local ToE was earlier and more widespread for frequency-based FWI metrics (FWI<sub>95d</sub> and FWI<sub>fwsl</sub>) than for magnitude-based metrics (FWI<sub>fs</sub> and FWI<sub>max</sub>). Heightened internal variability for annual maxima of other variables results in generally lower signal-to-noise detection (Diffenbaugh et al., 2017; King et al., 2015), and ambiguity in modeled changes in wind extremes concurrent with low fuel moistures that often result in peak FWI likely limit emergence as defined in this study (e.g., Tan, Chen, & Gan, 2018). Substantial geographic variability in ToE was evident with local emergence across much of the Mediterranean, southern Africa, southern Chile and Argentina, and Amazonia for most FWI metrics by 2030, consistent with the development of geographic hot spots of increasing drought risk imparted by anthropogenic climate change (e.g.,



**Figure 2.** Median ToE for FWl<sub>95d</sub> using (a) changes in all variables, (b) excluding monthly changes in precipitation totals, (c) excluding monthly changes in minimum relative humidity, (d) excluding monthly changes in maximum temperature, and (e) excluding monthly changes in wind speed. (f) The percent of global burnable land where median ToE occurs by date for the reference case and the four experiments.

Greve & Seneviratne, 2015; Gudmundsson & Seneviratne, 2016). By contrast, emergence was not found by 2050 in a majority of GCMs for much of equatorial Africa, portions of Australia, and sections of boreal systems in eastern Russia and Canada. A comparative analysis using bias-corrected GCM outputs showed similar results (Figures S1 and S2). We additionally visualize the number of GCMs with emergence commencing by 2019 (Figure S3), for which a majority of models depict emergence of FWI<sub>95d</sub> to be under way for approximately 22% of global burnable land surfaces.

Emergence is a function of the signal of change and persistence subject to internal climate variability as compared to the natural variability of the baseline period. We deconstruct the primary drivers of emergence in FWI metrics using two approaches. First, we present multimodel median changes in the four FWI metrics for the midtwenty-first century (2040–2069) relative to the baseline period (Figure S4). A majority of terrestrial surfaces show statistically significant increases in all metrics (t test; p < 0.05) for a majority of models by midcentury, including a doubling of  $FWI_{95d}$  versus the historical baseline for 31% of burnable global lands. For comparison, the multimodel median standard deviation calculated over the baseline period shows relatively high interannual variability in FWI metrics across portions of Australia which impede emergence based on a signal-to-noise ratio (Figure S5). Modeled changes in FWI metrics are broadly consistent with projected midtwenty-first century changes in the four meteorological variables used to calculate FWI during the core fire season (defined as the 17-model median date of maximum 90-day FWI over the baseline period), with substantial increases in tasmax and declines of greater than 2% in rhsmin across most land surfaces (Figure S6). Changes in pr during the core fire season exhibit more distinct geographic variability with decreases in the Mediterranean, Amazonia, and southern Africa where FWI signals emerge earliest, and increases across much of the high latitudes of North American and Eurasia where emergence is delayed. Tan, Gan, and Horton (2018) showed similar patterns for the ToE of consecutive dry days across the globe, with increases for most land surfaces except





**Figure 3.** (a) Time of emergence of anthropogenic climate change at subcontinental regions as depicted by the median ToE from 17 models. The four quadrants show the ToE for the frequency of FWI extremes (FWI<sub>95d</sub>; SE quadrant), length of the fire weather season (FWI<sub>fwsi</sub>; SW quadrant), peak 90-day FWI (FWI<sub>fs</sub>; NW quadrant), and annual maximum FWI (FWI<sub>max</sub>; NE quadrant). Light black lines show the geographic delineation of the regions. The lack of regional emergence before 2050 is denoted by dark gray. Time series of regional FWI metrics from 1861–2100 for 17 different GCMs of (b) FWI<sub>95d</sub> for the West North America region, (c) FWI<sub>95d</sub> for the Amazon region, (d) FWI<sub>fwsi</sub> for South Europe/Mediterranean region, and (e) FWI<sub>max</sub> for Alaska-Northwest Canada region. The bold black line shows the multimodel median 30-year moving average. ToE for individual models are denoted by red dots along the *x* axis. The 17-model median ToE is denoted by a black triangle for regions where at least nine models show emergence.

declines in tropical Africa and high latitudes. The Canadian Forest Fire Danger Rating system and thus FWI directly incorporate precipitation occurrence and dry spell lengths into measures of fuel moisture.

Second, we performed a sensitivity analysis to decompose the trends for each of the four meteorological variables used to calculate FWI had on ToE for FWI<sub>95d</sub> (supporting information). Results show that emergence is predominately imparted by *tasmax* increases. Calculations performed in the absence of modeled trends in *tasmax* resulted in 13.2% of the burnable lands with ToE by 2050, compared to 62% when all trends were considered (Figure 2). Regions where emergence occurs in the absence of increased *tasmax* are areas with notable decreases in *pr* and *rhsmin* during the fire season (Figure S6). Increased fire season *pr* across high latitudes







**Figure 4.** Amount of warming in annual global mean air temperature (1861–1910 baseline) at which anthropogenic signals emerge for (a) the frequency of days exceeding the 95th percentile ( $FWI_{95d}$ ), (b) the length of the fire weather season ( $FWI_{fwsl}$ ), (c) the peak 90-day average FWI ( $FWI_{fs}$ ), and (d) annual maximum FWI ( $FWI_{max}$ ). Maps show the 17-model median global temperature of emergence (GToE). Areas of light gray highlight where emergence does not occur by 2050. (e) The multimodel median proportion of burnable terrestrial surfaces for which emergence occurs as a function of global temperature increase beyond the preindustrial baseline.

of North America and Eurasia partially offsets the influence of warming trends on FWI per the sensitivity analysis by Flannigan et al. (2016), thus helping explain the delayed emergence of FWI metrics in those locations.

Regional emergence of FWI metrics was approximately 2 decades earlier than for local emergence (Figure 3a). Emergence occurs for FWI metrics in most regions by 2050, and modeled results suggest that a regional signal emerges by 2019 in many areas including the Mediterranean, portions of South America, and North America. The earlier and more robust ToE at regional scales versus local scales is congruent with previous ToE and broader detection and attribution studies pertaining to the reduction of internal variability through geographic aggregation (e.g., King et al., 2015). By contrast, regional emergence for a majority of models was not seen for any of the four metrics in northwestern North America, eastern Africa, and southern Asia.

We provide select examples of annualized time series of FWI metrics to illustrate the ToE framework and magnitude of projected changes in Figure 3. Models project an average of 3 times the baseline FWI<sub>95d</sub> by the end of the century for the western North America and Amazon regions (Figures 3b and 3c). Median ToE was 2018 and 1997 for these western North America and Amazon regions, respectively, although one model does not show emergence for the Amazon region (Figures S7–S10). Strong intermodel agreement of increased FWI<sub>fwsI</sub> was found for the Mediterranean region with all models showing emergence under way by 2030 and FWI<sub>fwsI</sub> being nearly twice that of baseline conditions by the end of the twenty-first century (Figure 3d). By contrast, fewer than half of the models show emergence of FWI<sub>max</sub> in the Alaska-Northwest Canada region (Figure 3e). With a 2 °C warming in global mean temperature over baseline, approximately 30 and 21% of global burnable land shows local emergence of FWI<sub>95d</sub> and FWI<sub>fwsI</sub>, respectively. Emergence at 2 °C warming is most pronounced in the Mediterranean, southern Africa, and portions of the Americas (Figures 4a–4d). With a 3 °C warming in global mean temperature, the geographic coverage of local emergence of FWI metrics nearly doubles beyond that associated with 2 °C warming (Figure 4e). The geographic extent of emergence for FWI<sub>fs</sub> and FWI<sub>max</sub> lags that of FWI<sub>95d</sub> and FWI<sub>fwsI</sub> with all metrics increasing commensurate with warming.

## 4. Discussion and Conclusion

Previous studies have detected increases in fire weather indices regionally and globally over a relatively short observational record (Abatzoglou & Kolden, 2013; Dowdy, 2017; Jolly et al., 2015). This study demonstrates an emergence of FWI-based metrics from "natural" variability for 33–62% of burnable lands globally by the midtwenty-first century due to anthropogenic climate change. Local emergence of the frequency of extreme fire weather days showed the most widespread changes, including modeled evidence of emergence commencing by 2019 for 22% of burnable lands globally. By contrast, emergence of magnitude-based metrics such as annual maxima was less robust and delayed relative to frequency-based metrics, suggesting distinct patterns in the character of fire weather changes resulting from anthropogenic forcing. Emergence for FWI is delayed and exhibits more geographic variability relative to the emergence of temperature-based measures (e.g., King et al., 2015) given the complementary (e.g., Mediterranean) and competing (e.g., boreal) roles of changes in precipitation during the fire season on FWI. The ToE framework does not explicitly exclude natural variability from influencing changes in FWI, but the persistence of time scales used in quantifying emergence, consideration of 17 climate models, and sensitivity to temperature as the predominant driver of projected changes in FWI combine to implicate anthropogenic factors.

Emergence at regional scales occurs earlier than at local scales as noted in previous ToE and attribution studies (Fischer & Knutti, 2014; King et al., 2015). Local emergence of fire weather metrics has pronounced impacts for ecological processes, but the regional emergence of fire weather extremes may be more perceivable from a societal perspective. We note that the spatial scale of our analysis is inadequate for resolving mesoscale processes important for local extreme fire weather conditions such as those that arise in southwestern California (e.g., Hughes & Hall, 2010). Heightened regional-scale fire weather and fuel aridity has been shown to have strong relationships with macroscale burned area (e.g., Williams & Abatzoglou, 2016). Increased regional fire weather contributes to geographic synchronicity in fire activity in flammability-limited regions that compromises regional and national resources available for suppression activities, thereby decreasing suppression efficacy for new ignitions (Flannigan, Stocks, et al., 2009) and require reallocating funds meant for preventative maintenance to fire suppression efforts (North et al., 2015). Likewise, the regional emergence in fire weather metrics is reasoned to have notable impacts on regional to continental air quality and human health (McClure & Jaffe, 2018; Spracklen et al., 2009).

The emergence of fire weather metrics viewed through the lens of changes in global mean temperature provides an alternative policy-based framework for viewing potential risks. We demonstrate that policy targets to limit warming of global mean temperature to no more than 2 °C above preindustrial levels have substantial implications for changes in the local FWI emergence. The geographic extent of emergence of the four FWI metrics doubles between a 2 and 3 °C warming. These results are in qualitative agreement to the findings of Turco et al. (2018), who showed that while modeled estimates of burned area in Mediterranean Europe increased with warming, the magnitude of such increases was substantially reduced when warming was held below 2 °C above preindustrial levels. Whereas observations suggest that changes have already occurred for aspects of fire weather metrics, these results highlight the risk avoidance for heightened fire weather conditions, fire potential, and associated fire hazards through climate mitigation efforts.

Projected changes in global fire activity broadly show increases in flammability-limited systems where fuel abundance in not limiting (Flannigan, Krawchuk, et al., 2009; Moritz et al., 2012). While this study does not explicitly attempt to model fire activity, our results generally support previous efforts given the strong contemporary relationships between FWI in both burned area in many flammability-limited regions globally (Abatzoglou et al., 2018) and the occurrence of extremely energetic fire events, many of which have had notable societal impacts (Bowman et al., 2017). Different fire weather indices each engender varied sensitivities to temperature, precipitation, humidity, and wind (e.g., Flannigan et al., 2016) and are reasoned to show

varied response to anthropogenic climate forcing (e.g., Abatzoglou & Williams, 2016). The FWI is not the only climate variable or biophysical property that influences fire activity or fuel dryness, and may not adequately capture the influence of live fuel moisture as it pertains to potential fire behavior (Ruffault et al., 2018). There are many reasons why changes in FWI should not directly translate into emergent changes in realized fire activity such as in fuel-limited fire regimes where FWI exhibits negligible influence on annual burned area (e.g., Abatzoglou et al., 2018), as well as those due to vegetation-fire feedback in a changing climate (e.g., Littell et al., 2018), and due to human activity altering climatic driven trends in fire activity through fire suppression and land use changes (Fréjaville & Curt, 2017; Marlon et al., 2012). Nonetheless, increased fire weather in a changing climate will be one of many ingredients that will continue to shape changes in global fire activity in the Anthropocene and will require local adaptation efforts to curb negative fire impacts.

#### Acknowledgments

Archive climate model output from CMIP5 can be accessed at https://esgfnode.llnl.gov/projects/cmip5/. J.T.A. was partially supported by the National Science Foundation under award DMS-1520873.

### References

Abatzoglou, J. T., & Kolden, C. A. (2013). Relationships between climate and macroscale area burned in the western United States. International Journal of Wildland Fire, 22(7), 1003–1020. https://doi.org/10.1071/WF13019

Abatzoglou, J. T., & Williams, A. P. (2016). Impact of anthropogenic climate change on wildfire across western US forests. Proceedings of the National Academy of Sciences, 113(42), 11,770–11,775. https://doi.org/10.1073/pnas.1607171113

- Abatzoglou, J. T., Williams, A. P., Boschetti, L., Zubkova, M., & Kolden, C. A. (2018). Global patterns of interannual climate-fire relationships. Global Change Biology, 24, 5164–5175. https://doi.org/10.1111/gcb.14405
- Andela, N., Morton, D. C., Giglio, L., Chen, Y., van der Werf, G. R., Kasibhatla, P. S., et al. (2017). A human-driven decline in global burned area. Science, 356(6345), 1356–1362. https://doi.org/10.1126/science.aal4108
- Arora, V. K., & Melton, J. R. (2018). Reduction in global area burned and wildfire emissions since 1930s enhances carbon uptake by land. *Nature Communications*, 9(1), 1326. https://doi.org/10.1038/s41467-018-03838-0

Barbero, R., Abatzoglou, J. T., Steel, E. A., & Larkin, N. K. (2014). Modeling very large-fire occurrences over the continental United States from weather and climate forcing. *Environmental Research Letters*, 9(12), 124009. https://doi.org/10.1088/1748-9326/9/12/124009

Barnett, T. P., Pierce, D. W., Hidalgo, H. G., Bonfils, C., Santer, B. D., Das, T., et al. (2008). Human-induced changes in the hydrology of the western United States. *Science*, 319(5866), 1080–1083. https://doi.org/10.1126/science.1152538

Bowman, D. M. J. S., Williamson, G. J. G., Abatzoglou, J. T., Kolden, C. A., Cochrane, M. A., & Smith, A. M. S. (2017). Human exposure and sensitivity to globally extreme wildfire events. *Nature Ecology & Evolution*, 1(3), 58. https://doi.org/10.1038/s41559-016-0058

Bradstock, R. A. (2010). A biogeographic model of fire regimes in Australia: Current and future implications. *Global Ecology and Biogeography*, 19(2), 145–158. https://doi.org/10.1111/j.1466-8238.2009.00512.x

- Cannon, A. J. (2018). Multivariate quantile mapping bias correction: An *N*-dimensional probability density function transform for climate model simulations of multiple variables. *Climate Dynamics*, 50(1–2), 31–49. https://doi.org/10.1007/s00382-017-3580-6
- Di Giuseppe, F., Pappenberger, F., Wetterhall, F., Krzeminski, B., Camia, A., Libertá, G., & San Miguel, J. (2016). The potential predictability of fire danger provided by numerical weather prediction. *Journal of Applied Meteorology and Climatology*, 55(11), 2469–2491. https://doi.org/10.1175/JAMC-D-15-0297.1
- Diffenbaugh, N. S., & Scherer, M. (2011). Observational and model evidence of global emergence of permanent, unprecedented heat in the 20th and 21st centuries. *Climatic Change*, 107(3–4), 615–624. https://doi.org/10.1007/s10584-011-0112-y
- Diffenbaugh, N. S., Singh, D., Mankin, J. S., Horton, D. E., Swain, D. L., Touma, D., et al. (2017). Quantifying the influence of global warming on unprecedented extreme climate events. *Proceedings of the National Academy of Sciences*, 114(19), 4881–4886. https://doi.org/10.1073/ pnas.1618082114
- Dowdy, A. J. (2017). Climatological variability of fire weather in Australia. Journal of Applied Meteorology and Climatology, 57(2), 221–234. https://doi.org/10.1175/JAMC-D-17-0167.1
- Field, R. D., Van Der Werf, G. R., Fanin, T., Fetzer, E. J., Fuller, R., Jethva, H., et al. (2016). Indonesian fire activity and smoke pollution in 2015 show persistent nonlinear sensitivity to El Niño-induced drought. *Proceedings of the National Academy of Sciences*, 113(33), 9204–9209. https://doi.org/10.1073/pnas.1524888113
- Fischer, E. M., & Knutti, R. (2014). Detection of spatially aggregated changes in temperature and precipitation extremes. *Geophysical Research Letters*, *41*, 547–554. https://doi.org/10.1002/2013GL058499
- Flannigan, M., Cantin, A. S., de Groot, W. J., Wotton, M., Newbery, A., & Gowman, L. M. (2013). Global wildland fire season severity in the 21st century. Forest Ecology and Management, 294, 54–61. https://doi.org/10.1016/j.foreco.2012.10.022
- Flannigan, M., Stocks, B., Turetsky, M., & Wotton, M. (2009). Impacts of climate change on fire activity and fire management in the circumboreal forest. *Global Change Biology*, 15(3), 549–560. https://doi.org/10.1111/j.1365-2486.2008.01660.x
- Flannigan, M. D., Krawchuk, M. A., de Groot, W. J., Wotton, B. M., & Gowman, L. M. (2009). Implications of changing climate for global wildland fire. International Journal of Wildland Fire, 18(5), 483–507. https://doi.org/10.1071/WF08187
- Flannigan, M. D., Wotton, B. M., Marshall, G. A., de Groot, W. J., Johnston, J., Jurko, N., & Cantin, A. S. (2016). Fuel moisture sensitivity to tem-

perature and precipitation: Climate change implications. Climatic Change, 134(1–2), 59–71. https://doi.org/10.1007/s10584-015-1521-0

- Fréjaville, T., & Curt, T. (2017). Seasonal changes in the human alteration of fire regimes beyond the climate forcing. *Environmental Research Letters*, *12*(3), 35006. https://doi.org/10.1088/1748-9326/aa5d23
- Friedl, M. A., Sulla-Menashe, D., Tan, B., Schneider, A., Ramankutty, N., Sibley, A., & Huang, X. (2010). MODIS collection 5 global land cover: Algorithm refinements and characterization of new datasets. *Remote Sensing of Environment*, 114(1), 168–182. https://doi.org/10.1016/j. rse.2009.08.016
- Gillett, N. P., Weaver, A. J., Zwiers, F. W., & Flannigan, M. D. (2004). Detecting the effect of climate change on Canadian forest fires. *Geophysical Research Letters*, 31, L18211. https://doi.org/10.1029/2004GL020876
- Greve, P., & Seneviratne, S. I. (2015). Assessment of future changes in water availability and aridity. *Geophysical Research Letters*, 42, 5493–5499. https://doi.org/10.1002/2015GL064127
- Groot, W. de Wotton, B., & Flannigan, M. (2014). Wildland fire danger rating and early warning systems. Wildfire Hazards, Risks, and Disasters. Gudmundsson, L., & Seneviratne, S. I. (2016). Anthropogenic climate change affects meteorological drought risk in Europe. *Environmental Research Letters*, *11*(4), 44005. https://doi.org/10.1088/1748-9326/11/4/044005

- Hawkins, E., Anderson, B., Diffenbaugh, N., Mahlstein, I., Betts, R., Hegerl, G., et al. (2014). Uncertainties in the timing of unprecedented climates. *Nature*, *511*(7507), E3–E5. https://doi.org/10.1038/nature13523
- Hawkins, E., & Sutton, R. (2012). Time of emergence of climate signals. *Geophysical Research Letters*, 39, L01702. https://doi.org/10.1029/2011GL050087
- Higuera, P. E., Abatzoglou, J. T., Littell, J. S., & Morgan, P. (2015). The changing strength and nature of fire-climate relationships in the northern Rocky Mountains, USA, 1902–2008. *PLoS One*, *10*(6), e0127563. https://doi.org/10.1371/journal.pone.0127563
- Hughes, M., & Hall, A. (2010). Local and synoptic mechanisms causing Southern California's Santa Ana winds. *Climate Dynamics*, 34(6), 847–857.
- Johnston, F. H., Henderson, S. B., Chen, Y., Randerson, J. T., Marlier, M., DeFries, R. S., et al. (2012). Estimated global mortality attributable to smoke from landscape fires. *Environmental Health Perspectives*, 120(5), 695–701. https://doi.org/10.1289/ehp.1104422
- Jolly, W., Cochrane, M., Freeborn, P., Holden, Z. A., Brown, T., Williamson, G. J., & Bowman, D. M. J. S. (2015). Climate-induced variations in global wildfire danger from 1979 to 2013. *Nature Communications*, 6(1), 7537. https://doi.org/10.1038/ncomms8537
- Kasischke, E. S., & Turetsky, M. R. (2006). Recent changes in the fire regime across the North American boreal region—Spatial and temporal patterns of burning across Canada and Alaska. *Geophysical Research Letters*, 33, L09703. https://doi.org/10.1029/2006GL025677
- King, A. D., Donat, M. G., Fischer, E. M., Hawkins, E., Alexander, L. V., Karoly, D. J., et al. (2015). The timing of anthropogenic emergence in simulated climate extremes. *Environmental Research Letters*, 10(9), 94015. https://doi.org/10.1088/1748-9326/10/9/094015
- Kirchmeier-Young, M. C., Zwiers, F. W., Gillett, N. P., & Cannon, A. J. (2017). Attributing extreme fire risk in Western Canada to human emissions. *Climatic Change*, 144(2), 365–379. https://doi.org/10.1007/s10584-017-2030-0
- Knorr, W., Jiang, L., & Arneth, A. (2016). Climate, CO<sub>2</sub> and human population impacts on global wildfire emissions. *Biogeosciences*, *13*(1), 267–282. https://doi.org/10.5194/bg-13-267-2016
- Lehner, F., Deser, C., & Terray, L. (2017). Toward a new estimate of "time of emergence" of anthropogenic warming: Insights from dynamical adjustment and a large initial-condition model ensemble. *Journal of Climate*, 30(19), 7739–7756. https://doi.org/10.1175/JCLI-D-16-0792.1
- Li, H., Sheffield, J., & Wood, E. F. (2010). Bias correction of monthly precipitation and temperature fields from intergovernmental panel on climate change AR4 models using equidistant quantile matching. *Journal of Geophysical Research*, *115*, D10101. https://doi.org/10.1029/2009JD012882
- Littell, J. S., McKenzie, D., Wan, H. Y., & Cushman, S. A. (2018). Climate change and future wildfire in the western United States: An ecological approach to nonstationarity. *Earth's Future*, 6(8), 1097–1111. https://doi.org/10.1029/2018EF000878
- Lyu, K., Zhang, X., Church, J. A., Slangen, A. B. A., & Hu, J. (2014). Time of emergence for regional sea-level change. *Nature Climate Change*, 4(11), 1006–1010. https://doi.org/10.1038/nclimate2397
- Maraun, D. (2013). When will trends in European mean and heavy daily precipitation emerge? *Environmental Research Letters*, 8(1), 14004. https://doi.org/10.1088/1748-9326/8/1/014004
- Marlon, J. R., Bartlein, P. J., Gavin, D. G., Long, C. J., Anderson, R. S., Briles, C. E., et al. (2012). Long-term perspective on wildfires in the western USA. *Proceedings of the National Academy of Sciences*, 109(9), E535–E543. https://doi.org/10.1073/pnas.1112839109
- McClure, C. D., & Jaffe, D. A. (2018). US particulate matter air quality improves except in wildfire-prone areas. Proceedings of the National Academy of Sciences, 115(31), 7901–7906. https://doi.org/10.1073/pnas.1804353115
- Moritz, M. A., Parisien, M.-A., Batllori, E., Krawchuk, M. A., Van Dorn, J., Ganz, D. J., & Hayhoe, K. (2012). Climate change and disruptions to global fire activity. *Ecosphere*, 3(6), 1–22.
- North, M. P., Stephens, S. L., Collins, B. M., Agee, J. K., Aplet, G., Franklin, J. F., & Fulé, P. Z. (2015). Reform forest fire management. *Science*, 349(6254), 1280–1281. https://doi.org/10.1126/science.aab2356
- Pechony, O., & Shindell, D. T. (2010). Driving forces of global wildfires over the past millennium and the forthcoming century. Proceedings of the National Academy of Sciences of the United States of America, 107(45), 19167–19170. https://doi.org/10.1073/pnas.1003669107
- Pierce, D. W., Barnett, T. P., Hidalgo, H. G., Das, T., Bonfils, C., Santer, B. D., Bala, G., et al. (2008). Attribution of declining western US snowpack to human effects. *Journal of Climate*, 21(23), 6425–6444. https://doi.org/10.1175/2008JCLI2405.1
- Pierce, D. W., Cayan, D. R., Maurer, E. P., Abatzoglou, J. T., & Hegewisch, K. C. (2015). Improved Bias correction techniques for hydrological simulations of climate change\*. *Journal of Hydrometeorology*, 16(6), 2421–2442. https://doi.org/10.1175/JHM-D-14-0236.1
- Randerson, J. T., Chen, Y., Werf, G. R., Rogers, B. M., & Morton, D. C. (2012). Global burned area and biomass burning emissions from small fires. Journal of Geophysical Research, 117, G04012. https://doi.org/10.1029/2012JG002128
- Risser, M. D., & Wehner, M. F. (2017). Attributable human-induced changes in the likelihood and magnitude of the observed extreme precipitation during hurricane Harvey. *Geophysical Research Letters*, 44, 12,457–12,464. https://doi.org/10.1002/2017GL075888
- Rosenzweig, C., Karoly, D., Vicarelli, M., Neofotis, P., Wu, Q., Casassa, G., et al. (2008). Attributing physical and biological impacts to anthropogenic climate change. *Nature*, 453(7193), 353–357. https://doi.org/10.1038/nature06937
- Ruffault, J., Martin-StPaul, N., Pimont, F., & Dupuy, J.-L. (2018). How well do meteorological drought indices predict live fuel moisture content (LFMC)? An assessment for wildfire research and operations in Mediterranean ecosystems. *Agricultural and Forest Meteorology*, 262, 391–401. https://doi.org/10.1016/j.agrformet.2018.07.031
- Spracklen, D. V., Mickley, L. J., Logan, J. A., Hudman, R. C., Yevich, R., Flannigan, M. D., & Westerling, A. L. (2009). Impacts of climate change from 2000 to 2050 on wildfire activity and carbonaceous aerosol concentrations in the western United States. *Journal of Geophysical Research*, 114, D20301. https://doi.org/10.1029/2008JD010966
- Stott, P. A., Christidis, N., Otto, F. E. L., Sun, Y., Vanderlinden, J.-P., van Oldenborgh, G. J., et al. (2015). Attribution of extreme weather and climate-related events. *Wiley Interdisciplinary Reviews: Climate Change*, 7(1), 23–41. https://doi.org/10.1002/wcc.380
- Tan, X., Chen, S., & Gan, T. Y. (2018). Multi-model extreme event attribution of the weather conducive to the 2016 Fort McMurray wildfire. Agricultural and Forest Meteorology, 260, 109–117.
- Tan, X., Gan, T. Y., & Horton, D. E. (2018). Projected timing of perceivable changes in climate extremes for terrestrial and marine ecosystems. Global Change Biology, 24, 4696–4708. https://doi.org/10.1111/gcb.14329
- Turco, M., Rosa-Cánovas, J. J., Bedia, J., Jerez, S., Montávez, J. P., Llasat, M. C., & Provenzale, A. (2018). Exacerbated fires in Mediterranean Europe due to anthropogenic warming projected with non-stationary climate-fire models. *Nature Communications*, 9(1), 3821. https://doi. org/10.1038/s41467-018-06358-z
- Turetsky, M. R., Benscoter, B., Page, S., Rein, G., Van Der Werf, G. R., & Watts, A. (2015). Global vulnerability of peatlands to fire and carbon loss. *Nature Geoscience*, 8(1), 11–14. https://doi.org/10.1038/ngeo2325
- van der Werf, G. R., Randerson, J. T., Giglio, L., Gobron, N., & Dolman, A. J. (2008). Climate controls on the variability of fires in the tropics and subtropics. *Global Biogeochemical Cycles*, 22, GB3028. https://doi.org/10.1029/2007GB003122
- Van Wagner, C. E. (1987). Development and structure of the Canadian forest Fire Weather Index system. Can. For. Serv., Forestry Tech. Rep.

- Ward, D. S., Shevliakova, E., Malyshev, S., & Rabin, S. (2018). Trends and variability of global fire emissions due to historical anthropogenic activities. *Global Biogeochemical Cycles*, 32, 122–142. https://doi.org/10.1002/2017GB005787
- Westerling, A. (2016). Increasing western US forest wildfire activity: Sensitivity to changes in the timing of spring. *Philosophical Transactions* of the Royal Society, B: Biological Sciences, 371(1696), 20150178. https://doi.org/10.1098/rstb.2015.0178
- Williams, A. P., & Abatzoglou, J. T. (2016). Recent advances and remaining uncertainties in resolving past and future climate effects on global fire activity. *Current Climate Change Reports*, 2(1), 1–14. https://doi.org/10.1007/s40641-016-0031-0
- Williams, A. P., Seager, R., Abatzoglou, J. T., Cook, B. I., Smerdon, J. E., & Cook, E. R. (2015). Contribution of anthropogenic warming to California drought during 2012–2014. Geophysical Research Letters, 42, 6819–6828. https://doi.org/10.1002/2015GL064924